Author’s response to reviews

Title: Earliella scabrosa-associated Postoperative Endophthalmitis after Phacoemulsification with Intraocular Lens Implantation: a Case Report

Authors:

Hong He (32841542@qq.com)
Xiaolian Chen (xiaoch88@126.com)
Hongshan Liu (2034870359@qq.com)
Jiaochan Wu (512165286@qq.com)
Xingwu Zhong (zhongxwu@mail.sysu.edu.cn)

Version: 3 Date: 25 Jan 2017

Author’s response to reviews:

Dear editors and reviewers,

Thank you for your letter and for the reviewers’ comments on our manuscript entitled “The First Case Report of Earliella scabrosa-associated Postoperative Endophthalmitis after Phacoemulsification with Intraocular Lens Implantation” (ID: [BOPH-D-17-00446]). All of these comments were very helpful to use while we revised and improved our paper. We have carefully studied these comments and made corrections that we hope will meet with your approval. The responses to the reviewers’ comments are provided below.

Reviewer’s Comment: 1. we not that you have not included any declarations in your manuscript. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.

Abbreviations

Ethics approval and consent to participate

Consent for publication

Availability of data and material

Competing interests

Funding
Authors' contributions

Acknowledgements

Response: In consideration of the Reviewer’s suggestion, all the sections including abbreviations, Ethics approval and consent to participate, Consent for publication, Availability of data and material, Competing interests, Funding, Authors’ contributions and Acknowledgements were added under the heading “Declarations”.

We would like to express our great appreciation to you and the reviewers for the comments you have provided regarding our paper. If you have any further queries, please do not hesitate to contact us.

Kind regards,

Hong He

Dear editors and reviewers,

Thank you for your letter and for the reviewers’ comments on our manuscript entitled “The First Case Report of Earliella scabrosa-associated Postoperative Endophthalmitis after Phacoemulsification with Intraocular Lens Implantation” (ID: [BOPH-D-17-00446]). All of these comments were very helpful to use while we revised and improved our paper. We have carefully studied these comments and made corrections that we hope will meet with your approval. The responses to the reviewers’ comments are provided below.

Reviewer’s Comment: 1. we not that you have not included any declarations in your manuscript. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.

Abbreviations

Ethics approval and consent to participate

Consent for publication

Availability of data and material

Competing interests

Funding
Authors' contributions

Acknowledgements

Response: In consideration of the Reviewer’s suggestion, all the sections including abbreviations, Ethics approval and consent to participate, Consent for publication, Availability of data and material, Competing interests, Funding, Authors’ contributions and Acknowledgements were added under the heading “Declarations”.

We would like to express our great appreciation to you and the reviewers for the comments you have provided regarding our paper. If you have any further queries, please do not hesitate to contact us.

Kind regards,

Hong He